Author's response to reviews

Title: Plasma branched-chain and aromatic amino acid concentration after ingestion of an urban or rural diet in rural Mexican women.

Authors:

Adriana M. López (adrimar24@gmail.com)
Lilia G. Noriega (lnoriegal@gmail.com)
Margarita Diaz (maggiediazm@yahoo.com)
Nimbe Torres (nimbester@gmail.com)
Armando R. Tovar (tovar.ar@gmail.com)

Version: 6  Date: 20 January 2015

Author's response to reviews:

Please find enclosed a new version of our manuscript where we have removed the final sentences in abstract and conclusions as previously requested by the reviewer. We haven’t rephrased the sentence since the non-causal relationship between BCAA and obesity/diabetes has already been addressed in the discussion.

We hope that this new version will qualify our work for publication in BMC Obesity.